Author’s response to reviews

Title: Drug-related problems and medication reviews among old people with dementia

Authors:

Bettina Pfister (bettina.pfister@vll.se)
Jeanette Jonsson (jeanette.jonsson@vll.se)
Maria Gustafsson (maria.gustafsson@umu.se)

Version: 2 Date: 24 Oct 2016

Author’s response to reviews:

Thank you for your comments! We have made a second review taking into account the reviewers’ comments.

Reviewer reports:

Can you please reformat the manuscript with the Tables at the end of the manuscript and not in the middle of the main text of the body of the paper please. Can you also please check the formatting of some of the references in the reference list, some of them appear to be incomplete like number 2 and 13.

Thank you for your comments. All tables are now at the end of the manuscript and not in the middle. All references have been checked and the ones that were incomplete have been changed. However, I have not been able to change reference number 13. I have looked at other papers using this reference, and the format is the same as in the present paper. I hope that it is ok.

Yours sincerely

Maria Gustafsson